Author's response to reviews

Title: An alternative to the hand searching gold standard: Validating methodological search filters using relative recall

Authors:

Margaret Sampson (msampson@uottawa.ca)
Li Zhang (li.zhang@usask.ca)
Andra Morrison (andram@ccohta.ca)
Nicholas J Barrowman (nbarrowman@cheo.on.ca)
Tammy J Clifford (tammyc@ccohta.ca)
Robert W Platt (rplatt@po-box.mcgill.ca)
Terry P Klassen (terry.klassen@ualberta.ca)
Daivd Moher (dmoher@uottawa.ca)

Version: 2 Date: 29 March 2006

Author's response to reviews:

There was an empty figure box on page 9 of the manuscript. Prior to submission we moved Figure 1 to the end of the manuscript, but the box was left behind. The empty box has been deleted. The reference manager ID number from the beginning of Appendix 2 have also been deleted. They should have been deleted prior to submission. No information was added and no other changes were made.

Apologies for the inconvenience.